Author’s response to reviews

Title: Sleep Quality in Middle-aged and Elderly Chinese: distribution, associated factors and associations with cardio-metabolic risk factors

Authors:

Nazanin Haseli-Mashhadi (hasseli57@yahoo.com)
Tony Dadd (tony.dadd@unilever.com)
An Pan (apan@sibs.ac.cn)
Zhijie Yu (zjiyu@sibs.ac.cn)
Xu Lin (xlin@sibs.ac.cn)
Oscar H Franco (o.H.Franco@warwick.ac.uk)

Version: 4 Date: 22 April 2009

Author’s response to reviews: see over
Title: Sleep Quality in Middle-aged and Elderly Chinese: distribution, associated factors and associations with cardio-metabolic risk factors
Ms. Ref. No.: 5454064022443418

Editor BMC Public Health

Subject: re-submission of manuscript

Dear Editor,

Thank you for your interest in our paper: ‘Sleep Quality in Middle-aged and Elderly Chinese: distribution, associated factors and associations with cardio-metabolic risk factors’ by Nazanin Haseli-Mashhadi, Tony Dadd, An Pan, Zhijie Yu, Xu Lin and Oscar H. Franco and for accepting to publish it.

We considered carefully all the editorial comments, which were very useful to further improve our paper for BMC Public Health. The revised manuscript with further modifications according to the editorial comments has been uploaded in your submission system together with this letter.

We hope this new version of our manuscript is now suitable for publication in BMC Public Health.

Yours truly,
(On behalf of the authors)

Dr Oscar H. Franco MD DSc PhD FESC
Assistant Clinical Professor in Public Health
B-160, Health Sciences Research Institute
University of Warwick
Coventry, CV4 7AL
United Kingdom
email: O.H.Franco@warwick.ac.uk
phone: +44 (0) 2476 575884
mobile: +44 (0) 7545 003927